Author’s response to reviews

**Title:** Quality of antenatal care services at public health facilities of Bahir-Dar Special Zone, Northwest Ethiopia.

**Authors:**

- Tadese E Tafere Mr (tade_et@yahoo.com)
- Mirkuzie W Terie Dr (mirkuzie@yahoo.com)
- Yibeltal K Alemayehu Mr (kibeltal@gmail.com)

**Version:** 2 **Date:** 12 June 2013

**Author’s response to reviews:**

Quality of antenatal care services at public health facilities of Bahir-Dar Special Zone, Northwest Ethiopia

Tadese Ejigu (BSc, MPH) 1*, Mirkuzie Woldie (MD, MPH) 2, Yibeltal Kifle (BSc, MSc) 3

1 Department of public health, Bahir-Dar University, Bahir Dar, Ethiopia

2 Department of Health planning and management, Jimma university, Jimma Ethiopia

3 Department of Health planning and management, Jimma university, Jimma Ethiopia

E-mail: Tadese Ejigu: tade_et@yahoo.com; Mirkuzie Woldie: mirkuzie@yahoo.com;

Yibeltal Kifle: kibeltal@gmail.com

* Corresponding author